Author's response to reviews

Title: Banded Gastric Bypass - four years follow up in a prospective multicenter analysis

Authors:

Lemmens Luc (luc.lemmens@telenet.be)
Karcz Konrad (sekretariat@drkarcz.de)
Fink Jodok (jodok.fink@uniklinik-freiburg.de)
Bukhari Waleed (wldbukhari@yahoo.com.uk)
Simon Kuesters (simon.kuesters@uniklinik-freiburg.de)

Version: 4 Date: 11 October 2014

Author's response to reviews:

Dear editors,

the remarks concerning ethics approval and informed consent were added.

Figure 1a was omitted since it was already published.

email addresses off all authors were included on the title page